Author’s response to reviews

Title: Company size, work-home interference, and well-being of self-employed entrepreneurs

Authors:

Isabelle Godin (igodin@ulb.ac.be)

Pierre Desmarez (pdesmar@ulb.ac.be)

Céline Mahieu (cemahieu@ulb.ac.be)

Version: 2 Date: 31 Oct 2017

Author’s response to reviews:

1. Table 1: split between entrepreneurs characteristics and workplace characteristics is made clear.
2. Tables' titles have been changed according to reviewer's proposition. Those changes are highlighted in the text in green.